Author's response to reviews

Title: The influence of family history of Hypertension on disease prevalence and associated metabolic risk factors among Sri Lankan adults

Authors:

Priyanga Ranasinghe (priyanga.ranasinghe@gmail.com)
Dilini N Cooray (dilininc@gmail.com)
Ranil Jayawardena (ranil7@gmail.com)
Prasad Katulanda (pkatulanda@yahoo.co)

Version: 2 Date: 11 March 2015

Author's response to reviews:

Page and line numbers included
Acknowledgements - None